\begin{table}%t1
\caption{\label{table:properties-midisurvey}Object properties derived in previous measurements or in this study.} 
%\centerline
 {\small \begin{tabular}{lllllllllc}
\hline\hline\noalign{\smallskip}
Object (no.) & RA [h~m~s] & Dec. [$^{\circ}$\ \arcmin\ \arcsec] &
$d$ [pc] & $A_{\rm V}$ [mag] & SpTyp & $M_\star$ [$M_{\rm \odot}$] & $T_\star$ [K] & $L_\star$ [$L_{\rm \odot}$] & Age [Myr] \\
\hline
DR~Tau (I) & & & & & & & & $1.7_{\rm (det.)}$ & \\[1.0ex] 
DR~Tau (II) & $04\ 47\ 06.2_{\rm (2)}$ & $+16\ 58\ 43_{\rm 
(2)}$ & $140_{\rm (ass.)}$ & $1.6_{\rm (3)}$ & K$7_{\rm (3,4)}$ &
$0.80_{\rm (4)}$ & $4000_{\rm (4)}$ & $0.9_{\rm (3)}$ &
$3_{\rm (5)}$ \\[1.0ex] 
DR~Tau (III) & & & & & & & & $0.9_{\rm (3)}$ & \\[1.0ex] 
GW~Ori (I) & \raisebox{-1.5ex}{$05\ 29\ 08.4_{\rm (1)}$} &
\raisebox{-1.5ex}{$+11\ 52\ 13_{\rm (1)}$} &
\raisebox{-1.5ex}{$440_{\rm (6)}$} &
\raisebox{-1.5ex}{$1.3_{\rm (6)}$} &
\raisebox{-1.5ex}{G$0_{\rm (6)}$} &
\raisebox{-1.5ex}{$3.7_{\rm (6)}$} &   
\raisebox{-1.5ex}{$6000_{\rm (6)}$} & $40_{\rm (det.)}$ &
\raisebox{-1.5ex}{$1_{\rm (6)}$}\\[1.0ex] 
GW~Ori (II) & & & & & & & & $62_{\rm (6)}$ & \\[1.0ex]
HD~72106~B & $08\ 29\ 34.9_{\rm (1)}$ & $-38\ 36\ 21_{\rm 
(1)}$ & $290_{\rm  (9)}$ & $0.0_{\rm  (7)}$ & A$0_{\rm  (9)}$ & $1.8_{\rm  (8)}$ & $9500_{\rm  (7)}$ & $28_{\rm  (det.)}$ &  $10_{\rm  (8)}$ \\[1.0ex]
RU~Lup & $15\ 56\ 42.3_{\rm  (1)}$ & $-37\ 49\ 16_{\rm 
(1)}$ & $127_{\rm  (11)}$ & $0.50_{\rm  (11)}$ & K$8_{\rm  (11)}$ & $0.80_{\rm  (11)}$ & $4000_{\rm  (11)}$ & $1.3_{\rm  (11)}$ & $1_{\rm  (11)}$ \\[1.0ex]
HBC~639 & $16\ 26\ 23.4_{\rm  (2)}$ & $-24\ 20\ 60_{\rm  (2)}$ & $170_{\rm  (11)}$ & $5.7_{\rm  (11)}$ & K$0_{\rm  (11)}$ & $2.0_{\rm  (11)}$ & $4800_{\rm  (det.)}$ &  
$8.5_{\rm  (11)}$ & $2_{\rm  (11)}$ \\[1.0ex]
S~CrA~N (I,II) & \raisebox{-1.5ex}{$19\ 01\ 08.6_{\rm  (2)}$} &
\raisebox{-1.5ex}{$-36\ 57\ 20_{\rm  (2)}$} &
\raisebox{-1.5ex}{$130_{\rm  (12,13)}$} & $1.5_{\rm  (12)}$ &
K$3_{\rm  (12)}$ & $1.5_{\rm  (12)}$ & $4400_{\rm  (12)}$ &
$2.3_{\rm  (12)}$ & $3_{\rm  (12)}$ \\[1.0ex]
 S~CrA~S & & & & $1.0_{\rm  (12)}$ & M$0_{\rm  (12)}$
& $0.60_{\rm  (12)}$ & $3800_{\rm  (12)}$ & $1.0_{\rm  (det.)}$ & $1_{\rm  (12)}$ \\  
        \hline  
      \end{tabular} }
\medskip
References: $^{(1)}$:~Perryman~(\cite{perryman});  $^{(2)}$:~2~MASS catalogue~(Cutri et~al. \cite{cutri});  $^{(3)}$:~Muzerolle et~al. (\cite{muzerolle}); $^{(4)}$:~Mohanty et~al. (\cite{mohanty}); $^{(5)}$:~Greaves (\cite{greaves});  $^{(6)}$:~Calvet et~al. (\cite{calvet}); $^{(7)}$:~Sch\"utz et~al. (\cite{schuetz}); $^{(8)}$:~Wade et~al. (\cite{wade});  $^{(9)}$:~Vieira et~al. (\cite{vieira}); $^{(10)}$:~Folsom (\cite{folsom}); $^{(11)}$:~Gras-Vel\'azquez \& Ray (\cite{gras-velazquez}); $^{(12)}$:~Prato
et~al. (\cite{prato}); $^{(13)}$:~Johns-Krull et~al. (\cite{johns-krullII}).
\end{table}